

## **Author's response to reviews**

**Title:** Understanding Type 2 Diabetes Mellitus Screening Practices Among Primary Care Physicians: A Qualitative Chart-Stimulated Recall Study

### **Authors:**

Dina Hafez (dhafez@med.umich.edu)

Daniel Nelson (dannel@med.umich.edu)

Evan Martin (marteva@med.umich.edu)

Alicia Cohen (calicia@med.umich.edu)

Rebecca Northway (rnorthwa@med.umich.edu)

Jeffrey Kullgren (jkullgre@med.umich.edu)

**Version: 2 Date:** 17 Mar 2017

### **Author's response to reviews:**

Please make the following revisions to your manuscript:

1. Thank you for submitting an Authors' contribution section. We had noted the contributions of Alicia J Cohen and Rebecca Northway were missing. Please add their contributions to this section so that they can qualify as authors.

- We added the contributions of Drs. Cohen and Northway.

2. Please change the text under Consent to Publish to "not applicable".

- We made this change.

3. Please upload Appendix 1 as a supplementary file.

- We saved the Appendix as a separate file.

4. We would appreciate if you add a section titled “Supplementary files” where you list the following information about your supplementary material:

- File name
- Title of data
- Description of data

We added this section after the Conclusion.

Please ensure that all additional files are explicitly referred to in the main text. Any items which do not meet these requirements may be deleted by our production department.

- We refer to all Tables and the Appendix in our manuscript.

5. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

- We have uploaded only a clean version of our manuscript.